Author's response to reviews

Title: Assessment of the prevalence of intestinal parasitos and the associated risk factors among primary school children in Chencha town, southern Ethiopia.

Authors:

Ashenafi A Abossie L (asaboet@yahoo.com)
Mohammed S Seid I (mohamedseid2005@gmail.com)

Version: 2
Date: 21 October 2013

Author's response to reviews:

we have intended to show the problem intestinal parasite infection in this particular area since the problem is ignored by the zone health department and other concerned bodies. the results also seek the attention of other researchers and institutes to do more in this public health problem.